Author's response to reviews

Title: First evidence of a large CHEK2 duplication involved in cancer predisposition in an Italian family with hereditary breast cancer

Authors:

Gianluca Tedaldi (g.tedaldi@irst.emr.it)  
Rita Danesi (rita.danesi@irst.emr.it)  
Valentina Zampiga (valentina.zampiga@irst.emr.it)  
Michela Tebaldi (michela.tebaldi@irst.emr.it)  
Lucia Bedei (l.bedei@ausl.fo.it)  
Wainer Zoli (wainer.zoli@irst.emr.it)  
Dino Amadori (dino.amadori@irst.emr.it)  
Fabio Falcini (fabio.falcini@irst.emr.it)  
Daniele Calistri (daniele.calistri@irst.emr.it)

Version: 3  
Date: 17 June 2014

Author's response to reviews:

17.06.14

Ms. Cherry Battad

Journal Editorial Office  
BioMed Central  
Floor 6, 236 Gray's Inn Road  
London  
WC1X 8HB  
UK

Dear Ms. Battad,

Re: MS 1426338930111876

First evidence of a large CHEK2 duplication involved in cancer predisposition in
an Italian family with hereditary breast cancer by Gianluca Tedaldi et al.

We are returning the above manuscript, revised in accordance with the Editors’ request for an Ethics statement to be added. The information has been inserted in red font on page 6, lines 2-6.

We look forward to hearing from you.

Yours sincerely,

Gianluca Tedaldi, BSc

Corresponding Author